Author's response to reviews

Title: Meropenem vs standard of care for treatment of late onset sepsis in children of less than 90 days of age: study protocol for a randomised controlled trial

Authors:

Irja Lutsar (irja.lutsar@ut.ee)
Ursula MT Trafojer (trafojer@pediatria.unipd.it)
Paul T Heath (pheath@sgul.ac.uk)
Tuuli Metsvaht (tuuli.metsvaht@kliinikum.ee)
Joseph Standing (joseph.standing@nhs.net)
Susanna Esposito (susanna.esposito@unimi.it)
Vincent Meiffredi de Cabre (vincent.meiffredi@inserm.fr)
Clarissa Oeser (coeser@sgul.ac.uk)
Jean-Pierre Aboulker (jean-pierre.aboulker@inserm.fr)

Version: 6 Date: 28 September 2011

Author's response to reviews:

The addresses of all authors have been added on the title page of the manuscript. I was unable to find any typographical errors.